

Junior Management Science

journal homepage: www.jums.academy



Impact of Weather on the Stock Market Returns of Different Industries in Germany

Astrid Schulte-Huermann

WHU - Otto Beisheim School of Management

Abstract

Weather affects people's mood, according to psychological studies. For example, low temperature can cause aggression, whereas high temperature can induce apathy. Therefore, it may be possible that weather-influenced mood, driven by mood's impact on decision-making, exerts an influence on investment decisions and risk-taking behaviour. Thereby, it might also impact stock market returns. I examine market returns of nine industries in Germany. Empirical results illustrate two findings. First, a statistically significant, negative correlation between market returns and temperature, second, a different effect of weather on industrial sectors is identified. A significant correlation can be found for six out of nine sectors.

Keywords: Stock markets; investor behaviour; weather effect; market returns; decision-making.

1. Introduction

The impact of climate change has become noticeable on a global level, and Germany has experienced more extreme weather phenomena during the last years. According to the German weather service (Deutscher Wetterdienst (DWD)), the summer of 2018 marks the second hottest summer on record since 1881 (Imbery et al., 2018). Further, weather and climate implications on companies' earnings are increasingly discussed on executive earnings calls, and rating agencies have started to focus on weather implications on companies' economic performance (Williams et al., 2018).

Investigating weather implications on the economy is a relevant field of study that is already supported by extensive research. For example, weather has been linked to stock market performance. The underlying reason for connecting these fields is that research has found evidence that weather conditions affect mood and spawn mood misattribution. A change in emotional state can trigger an alternated risk assessment and in turn a revised investment behaviour. This can ultimately lead to changes in stock market returns. Based on this finding, my study analyses if a relationship between weather and stock market returns exists.

Several researchers have carried out studies in the field of weather implications on psychology. Howarth and Hoffman (1984) observed that human performance is positively correlated with sunshine, and aggressiveness increases with low temperature. On the opposite, high temperatures have been associated with apathy (Cao and Wei, 2005). On sunny days,

DOI: https://doi.org/10.5282/jums/v5i3pp295-311

people are inclined to rate their life satisfaction higher than on cloudy days (Schwarz, 1990). People's mood is linked to their decision-making (Schwarz, 1990). Thus, it is also a component of their investment decisions.

Turning to empirical research, Saunders (1993) was among the first who linked weather to investors' behaviour. He reported a negative correlation between NYSE index returns and the degree of cloudiness. His results support the conjecture that investors are overoptimistic on sunny days and more pessimistic on cloudy days. Further evidence for the negative relation between cloud cover and stock market returns were given by Hirshleifer and Shumway (2003). In 18 out of 26 cities, a negative relationship could be reported. Other research produced evidence for a negative correlation between stock returns and temperature (Cao and Wei, 2002; Chang et al., 2006; Kang et al., 2010).

Moreover, the seasonal affective disorder (SAD) has an impact on stock returns. Affected people show depressive symptoms during winter, and consequently, stock market returns are lower during this time of the year. Kamstra et al. (2003) call this effect winter blues.

1.1. Relevance

My paper expands on the existing research by examining the probable linkage between stock returns and weather variables temperature and precipitation with a focus on Germany and differentiating between sectors. The former is advocated to be among the most essential meteorological variables to alter people's mood. The latter is only included in a minor part of literature yet and therefore will be further explored. In particular, my analysis examines the impact of temperature and precipitation on the stock market return of different industries in Germany. Hence, I will contribute to a yet unexplored field by differentiating stock market return behaviour between sectors. Hitherto, only stock market indices on a global scale or across industries have been analysed. While most research focuses on adjusting the independent variable set containing weather data, refinements of the dependent variable in the form of separation of industries are somewhat limited.

Weather does not only impact mood and decision-making but also it has economic implications on companies' economic efficiency. Therefore, my study expands upon the relationship between weather and stock market returns by also incorporating economic implications. It is evident that the climate and weather implications on the economy might vary across sectors. This is because integral parts of the supply chain or the use of the finished product depend to ranging degrees on environmental circumstances as the weather. Temperature, for example, might influence the demand for some products. Especially in the Retail and Utilities industry, demand may increase during periods of lower temperatures, whereas the economic performance of other companies is harmed when temperatures are high. These effects might also be present in the opposite direction because high and low temperatures can both be beneficial and unfavourable, depending on the industry. I also conjecture that precipitation might carry weight in the determination of stock market returns.

Germany as the country of research has only been investigated once on a national level by Krämer and Runde (1997). Their study concluded that no systematic relationship exists for the German stock market returns and weather variables. The rejection of the null hypothesis is rather determined by the manner the hypothesis is phrased. My study will elucidate whether those outcomes can be confined and can be replicated for a later period of time.

As psychological literature has outlined, weather influences people's mood and behaviour. Consequently, the risk assessment and investment decisions of traders could be altered. If this holds, trader's evaluation of companies might be over- or under-optimistic depending on the weather. Likewise, hot temperatures might induce a change in the evaluation of some industries. Another approach could be that changes in weather influence the mood of traders only for some particular industries. Certain sector-returns might not be affected by the weather at all. The inference seems appropriate, if one considers the impact of changing weather conditions through global warming on politics and industries today. By dividing into industry groups, it is finally possible to assess whether all company returns are affected by the weather or if this phenomenon is limited to specific sectors.

From a psychological point of view, potential findings could be used for weather-based trading strategies. From an economic standpoint, a sector might be attractive for stock trading, if its market returns are influenced by weather. Industries that are not correlated to weather variables might be more valuable for a buy and hold strategy. In general, the findings could not only be interesting for traders, but also be interesting for investors, and financial managers.

1.2. Structure

The remaining part of this paper is structured as follows. Previous qualitative and quantitative work is discussed in section two. Section three outlines the methodology and section four describes the data set which was collected for both weather and stock market returns, including descriptive statistics. Section five outlines the underlying hypotheses and leads to the central part of this paper. A detailed analysis of the relationship of weather with stock market returns is given in section six by using both a bin-test and an EGARCH model. Section seven will discuss my findings. Section eight will provide a further outlook and highlights the potential limitation of my work; afterwards, section nine concludes.

2. Literature review

Thorough research on the topic has made it evident that behavioural finance uses a systematic approach for the analysis of the potential impact of weather on an investors' psychology. The first part, thus, will elaborate on the effects of weather on human beings' mood. The second part will discuss the impact of mood on decision-making, and the third part presents research that links weather conditions with investment behaviour and consequently stock markets. The underlying process is illustrated in Figure 1.

2.1. Weather and mood

One-third of humans respond sensitively to weather, which is characterised by changes in mood and human wellbeing (Kals, 1982). More precisely, weather can provoke compound physical and psychological responses, including changes in performance, aggressive behaviour, and interpersonal interactions (Lu and Chou, 2012; Shu and Hung, 2009). The phenomenon has been discussed in a large body of literature, where the weather has been linked with human behaviour (Hirshleifer and Shumway, 2003; Howarth and Hoffman, 1984; Parsons, 2001; Pilcher et al., 2002; Rind, 1996; Watson, 2000). In current research, temperature is the most prominent influencing factor for people's mood. Especially, low temperature can fortify aggressive behaviour and discomfort. This effect also holds for high temperatures where apathy and hysteria are caused in addition to aggression (Cao and Wei, 2005; Pilcher et al., 2002). Extreme temperatures can also induce a decline in task performance (Pilcher et al., 2002), as well as a greater willingness to offer help to other people (Cunningham, 1979). Sunshine, on the opposite, can lead to higher satisfaction on sunny days in comparison to cloudy days, and consequently can enhance people's optimism (Howarth and Hoffman, 1984). Moreover, sunshine has been proven to influence the amount of tipping, as confirmed by Rind (1996). On the contrary, lack



Figure 1: Self-created process model of the impact of external influences on mood and decision-making

of sunshine was noted to cause depression (Eagles, 1994) and suicide (Tietjen and Kripke, 1994). Sariannidis et al. (2016) claim that wind is directly linked with investors' feeling of safety. With higher wind speed, air pollution is diluted, and therefore particles of dirt accumulate less. Humidity is an indicator that can lower human comfort and simultaneously lead to more aggressive behaviour (Cao and Wei, 2005). Since ancient times, the impact of the moon, especially the full moon, has been associated with changes in human feelings. Research has confirmed a shift in the behaviour of people: higher occurrence of somnambulism, crime, and mental disorders (Zheng et al., 2001).

2.2. Mood and decision-making

As weather can affect mood in various ways, this part will review psychological evidence that a change in mood, feelings, or emotions can influence decision-making. Among scholars, no consensus on the definition of the three terms exists, and the words are frequently used interchangeably.

Schwarz was among the first to identify that mood influences decision-making (Schwarz, 1990). He constitutes that mood can have an impact on the judgement of the information capabilities of a person. Further, a good mood can induce a more positive appraisal of life satisfaction (Wright and Bower, 1992) and prospects (Hirshleifer and Shumway, 2003). Also, a more intense utilisation of simplifying heuristics by individuals was identified in the respective research (Bless et al., 1996). Consequently, individuals base their judgement on stereotypes and are more likely to identify a deviation of available information from pre-existing knowledge. Other research has found that positive mood can lead to a higher receptiveness to weak arguments (Mackie and Worth, 1991), while it can enhance problem-solving abilities and creativity (Isen, 2002). According to Schwarz (1990), not only mood but also feelings are linked to decisionmaking. He further claims that through misattribution, emotions are connected to a false origin and consequently, induce incorrect judgements. Thus, the capability to process information is hindered and especially investors falsely respond to irrelevant information in a state of happiness. Also, visceral factors exert an influence on human beings' preferences (Hirshleifer and Shumway, 2003). Summarising, mood affects judgement, choices, and decision-making.

2.3. Stock markets and weather

Behavioural Finance is a field of research that has as its central premise the investment decision of individuals. In contrast to Traditional Finance, it is based on the theory that decision-making by individuals is not always rational but based on the impact of emotions and mood. Kahneman and Tversky are considered essential contributors to Behavioural Finance, promoting the non-existence of a "homo economicus". Instead, cognitive biases and heuristics guide an individual's behaviour and lead to irrational decisions. Consequently, arguments of the field of Behavioural Finance support the view that investment decisions are linked to people's mood. As summarised in section 2.1, weather conditions can influence mood and can cause for instance misattribution. They might also lead to changes in risk assessment and consequently altered tendencies in investment activities. Pleasant weather can lead to more communication, which expands information transfer, thereby encouraging investment activity (Symeonidis et al., 2010). Favourable weather conditions can also induce good mood. Investors transfer this positive state and evaluate economic prospects more optimistic (Shu, 2010) as they underestimate risk and overestimate the probability of success (Wright and Bower, 1992). As a result, stock market returns might be influenced by weatherinduced effects. Existing research has not only linked mood to weather but also has investigated potential relationships between stock market returns and various meteorological variables. There is both literature supporting and rejecting the relationship between weather and stock returns.

Temperature (Cao and Wei, 2005; Chang et al., 2008; Floros, 2008; Keef and Roush, 2002), number of sunshine hours (Pardo and Valor, 2003; Shu and Hung, 2009), cloud cover (Hirshleifer and Shumway, 2003; Saunders, 1993) and air humidity levels (Chang et al., 2006; Pardo and Valor, 2003; Sariannidis et al., 2016) are, among others, the standard variables to measure weather's influence on stock market returns. Also, other nature-related variables such as the daylight-savings time change (Kamstra et al., 2000), the length of the night (Kamstra et al., 2003), and the lunar phases of the moon (Dichev and Janes, 2001; Zheng et al., 2001), have been used as weather proxies that might influence an investor's mood.

2.3.1. Evidence of the weather and stock market relationship

Saunders (1993) was among the first researchers to connect weather variables with stock market returns. By using daily returns of the Dow Jones Industrial Average from 1927 to 1989, as well as equal- and value-weighted daily percentage changes in the NYSE/AMEX index from 1962 to 1989, he identified a significant negative correlation between the percentage of cloud cover and stock returns. This effect is robust with respect to various market anomalies, including the Monday and January effect. Consequently, Saunders conjectures that the mood of investors is overoptimistic on sunny days.

Further evidence for this sunshine effect has been given by Hirshleifer and Shumway (2003) who access the level of cloudiness in 26 countries from 1982 to 1997. An OLS regression model and a logit model both indicate that cloudiness exhibits a strong negative effect on returns in the majority of countries. The results remain stable after controlling for seasonality. This finding emphasises the conjecture that overcast sky reinforces a downbeat mood. Therefore, the researchers raise the question of whether a trading strategy based on a sunshine effect exists.

The research of Dowling and Lucey (2005) emphasises the relationship between stock market returns and rain and concludes that both variables are negatively correlated. This outcome is in line with the study of Saunders (1993), as "85% of rain occurs on days with 100% cloud cover" (Dowling and Lucey, 2005, p. 18).

A vast base of stock-market-focused literature uses temperature as a weather proxy. For most of the studies, a negative relationship has been found between temperature and stock market returns: The higher the temperature, the smaller the return.

For example, in an investigation of eight stock exchanges, Cao and Wei (2002, 2005) have found a negative correlation between temperature and market returns.

Negative correlation of temperature, sunshine, and humidity on the returns of the Shanghai stock market has been documented (Kang et al., 2010). Similar results were identified by Yoon and Kang (2009) for Korean stock returns. Here, for the whole period of 1990 to 2006, high temperature and humidity negatively influenced stock returns, while high cloudiness and low humidity led to positive stock returns. For the Taiwan stock exchange, temperature and cloud cover have a significant effect while humidity is not significantly correlated (Chang et al., 2006).

Keef and Roush (2007, 2002) have studied both the Australian and the New Zealand Stock Exchange, deriving the conclusion that temperature negatively affects returns. Whereas wind speed and cloud cover do not have a significant impact on the Australian stock returns, wind is indeed significant for returns in New Zealand. A further study of the researchers from Keef and Roush (2005) does indicate a negative correlation between New Zealand stock prices and wind, as well as a positive effect of sunshine on bank bills.

Kamstra et al. (2000) draw attention to the effect of sleep desynchronosis, an interruption in sleep patterns, induced through daylight saving time changes on stock returns. Further, the seasonal affective disorder, caused by fewer daily sunshine hours, has a significant impact on stock returns in several countries (Hirshleifer and Shumway, 2003). The researchers provide evidence that longer nights correlate with lower stock returns.

Europe has also served as a ground for studies in the field of weather impact on stock exchanges (Floros, 2011, 2008; Sariannidis et al., 2016). Hereby, negative relationships between weather variables and stock returns were again identified (Shu and Hung, 2009).

2.3.2. Evidence against the weather and stock market relationship

Contradicting with previously discussed findings is the research by Trombley (1997) who replicated Saunders results without finding any significant correlations. Thus, he states that "[Saunders's] choice is the only comparison during this period that would produce a statistically significant test statistic and does not consider that the returns on the zero percent days are inconsistent with the existence of a weather effect (p.13)."

Replication of Saunders result with German data has also supported Trombley's view (Krämer and Runde, 1997). Further, the researchers Krämer and Runde claim, that the significance of results, and a systematic relationship between weather and stock returns, does mostly depend on the phrasing of the null hypothesis and the variables. Loughran and Schultz (2004) receive little evidence in their analysis of the effect of cloudiness in the location of the company's headquarter on the returns of NASDAQ stocks.

Moreover, wind speed is no indicator for returns, as proven by Shu and Hung (2009) who analysed 18 European countries from 1994 to 2004.

Extant work by Zhu and Goetzmann (2003), who focused on data of individual traders during the period 1992 to 1996, further proved that not the weather, but the bid-ask spread led to investors' propensity to buy or sell. These findings are consistent with Pardo and Valor (2003) who provide evidence that humidity is not correlated with the Madrid Stock Exchange. This serves as an example of rational human behaviour. Lu and Chou (2012) also derive that the relationship between weather and stock market returns is insignificant. However, they identified that humidity and wind exert a positive effect on the volatility of stock indices, as shown with the Shanghai Stock Exchange.

In contrast to the research of Dichev and Janes (2001), who identified a strong effect of the lunar cycle on stock returns that is consistent over the last 100 years and for the majority of the 24 countries analysed, Rotton and Rosenberg (1984) have gathered evidence that shows no significant effect of the lunar phases on the Dow Jones Index prices.

Taken together, the relationship between weather and stock returns is controversial (Floros, 2011; Sariannidis et al., 2016) and research results differ. Locations, data, and the time horizon used are distinct among extant research and cannot lead to a generalised conclusion. However, it can be said that there is a significant body of literature that has provided evidence on an existing relationship between weatherrelated variables and stock market returns. If that holds, the marginal investor's behaviour is dependent on the weather, the season of the year, the phases of the moon, the daylightsavings time, and the length of the night. According to Zhu and Goetzmann (2003), the market is therefore not efficient, and this leaves opportunities to exploit its inefficiency. Investors can benefit from the awareness of the impact of their mood, as Hirshleifer and Shumway (2003) Suggest. Nevertheless, they also claim that mood is influenced not only by weather but also by other factors and biases.

The existence of the weather's influence on stock returns cannot be denied and raises the question on whether this effect is limited to countries and the industry as a whole or whether sectors are affected differently by weather variables. Having a profound body of literature as a base, the material will be used as a guide to analyse a yet underexplored field of research.

3. Methodology

The purpose of this paper is to gain valuable insight into the impact of weather on stock market returns in different industries in Germany. Literature has made use of a variety of analysis tools to investigate the relationship between market returns and weather. For this purpose, weather variable data and industry return data have been collected. The following paragraphs will describe the methodology of my analysis which will be grouped into two types of tests. Afterwards, the data collection shall be discussed.

3.1. Bin test

The first analysis, following Saunders (1993), is called Bin test based on Cao and Wei (2005). For this, I will use both industry index returns and the meteorological variable temperature. It will enable the identification of relationships between temperature and returns by matching temperature and return data, grouping it into bins, and calculating z-scores to evaluate the statistical difference between the bins. However, this test cannot identify specific correlations and lacks the ability to control for known stock return anomalies. The underlying methodology and detailed steps will be explained in the following.

First, the daily return and respective temperature data are sorted in descending order and separated into four bins. For each group, the mean return and the frequency of positive return are calculated, and the mean returns of the first and the fourth bin, covering the lowest temperature and the highest temperature respectively, will be compared. This will allow for the evaluation of the potential significance in the difference in mean return. Another comparison will be made between the percentage of positive returns in the four bins to verify whether the return difference is caused by outliners. In detail, first, the four bins are identified by taking the range between maximum and minimum temperature and dividing by the total number of k bins with (k = 1, ..., 4).

$$\Delta = \left(\text{Temperature}_{\text{max}} - \text{Temperature}_{\text{min}}\right)/k \tag{1}$$

The bin ranges are then calculated as followed: $[Temp_{min} + \Delta]$, $[Temp_{min} + 2\Delta]$ and so forth. Based on the procedure by Saunders (1993), a z-test is performed to verify whether mean return differences of bin one and four are statistically significant.

$$z_{-} \operatorname{score}_{4,1}^{\operatorname{mean}} = \left(\mu_{4} - \mu_{1}\right) / \sqrt{\frac{\sigma_{4}^{2}}{n_{4}} + \frac{\sigma_{1}^{2}}{n_{1}}}$$
(2)

The variable μ_i describes mean return, σ_i^2 is variance of return and n_i the number of observations per bin *i* with (*i* = 1, or *k*). A z-statistic is also performed for the percentage of positive returns per bin, to examine whether there is a significant difference between frequency of positive returns in the two extreme bins.

$$z_{-}\operatorname{score}_{4,1}^{\operatorname{frequence}} = \left(p_4 - p_1\right) / \sqrt{\frac{p_4(1 - p_4)}{n_4} + \frac{p_1(1 - p_1)}{n_1}}$$
(3)

In this case, p_i is the percentage of positive return for bin i (i = 1, or k).

3.2. EGARCH model

In financial research, the generalised autoregressive conditional heteroscedasticity (GARCH) model has been used extensively to investigate the relation between weather variables and stock market returns. The Glosten, Jagannathan, and Runkle (GJR) GARCH model has been utilised by Chang et al. (2006) and Symeonidis et al. (2010), other researchers employed the autoregressive threshold GARCH (AR-TGARCH) (1,1) model (Floros, 2011) or a simple GARCH(1,1) analysis (Kang et al., 2010). My study will make use of an exponential GARCH (EGARCH) model by Nelson (1991), and it resembles the core of my thesis.

The EGARCH model is applied because it is frequently argued that stock return shocks are asymmetric. Positive return shocks have a lower impact on volatility than negative return shocks of the same magnitude. This phenomenon is called leverage effect. Further, the model captures volatility clustering, which is typical for financial time series. The omitted parameter restriction can result in more reliable optimisations. It also makes this model robust against changes in volatility and against long-run memory effects which are a result of time-varying clusters. Besides, the EGARCH model uses the natural logarithm ln(.) which guarantees the positivity of the parameters. As a result, the EGARCH model has been often seen to be superior to other models.

The regression equation of the EGARCH model including return *R*, prices *P*, and a stochastic error term, is expressed as follows:

$$R_{t,i} = b_0 + b_1 \sigma_t^2 R_{t-1,i} + b_2 \sigma_t^2 \operatorname{Temp}_t + b_3 \sigma_t^2 \operatorname{Prec}_t + \xi_t$$
 (4)

 $R_{t,j}$ is the daily industry return gain or loss for day t, calculated:

$$R_{t,j} = \ln(P_{t,j}) - \ln(P_{t-1,j})$$
(5)

For the industry j, the subscript $j \in$ Financials, Telecommunication, Technology, Consumer Non-Cyclicals, Industrials, Healthcare, Basic Materials, Consumer Cyclicals, Utilities. $R_t t - 1$), the lagged return variable, is incorporated to account for potential nonsynchronous trading effects. Temp_t is the daily average temperature at day t, Prec_t the precipitation height in mm at day t, and ε_t is the error term at day t. The formula is consistent across all industries. The same set of variables is used for all industries.

The variable $\sigma_{t,j}^2$ is the volatility of industry *j* at day *t* of the EGARCH (p,q) model. It is an estimate of the one period ahead variance including past information that is considered relevant. For this reason, it is called conditional variance. The formula for each single industry is given by:

$$\ln\left(\sigma_{t,j}^{2}\right) = \alpha_{0} + \sum_{l=1}^{p} \gamma_{l} \ln \sigma_{t-l,j}^{2} + \sum_{l=1}^{q} \alpha_{l} \left[\left| z_{t-l,j} \right| - E \left| z_{t-l,j} \right| \right] + \sum_{l=1}^{q} \xi_{l} z_{t-l,j}$$
(6)

The conditional variance consists of several parts: First, $Z_{t-l} = \frac{\varepsilon_{t-l}}{\sigma_{t-l}}$ is the standardized residual at time t-l and l describes the number of lags. γ_l measures the persistence in conditional volatility irrespective of any activity in the market. Second, α_l describes the GARCH effect, which represents the magnitude effect or the symmetry effect. It gives insights into the degree to which the size of shock influences the stock market response. Third, the parameter ξ_l measures the leverage effect, which takes into consideration the asymmetry, which is an advantage of the EGARCH model. For $\xi_l < 0$, positive shocks generate less volatility than negative shocks, which is evidence for the leverage effect. For asymmetric shocks, the variable ξ_l must be non-zero ($\xi_l \neq 0$).

Returns are not normal distributions, as the JB test in section 4.3 will prove, and standardized residuals are leptokurtic. For heavier tails, the generalized error distribution (GED) is a more appropriate distribution than the t-distribution and will offer more reliable test results. The GED has the density function:

$$f\left(\mu_{t},\sigma_{t}^{2},\nu\right) = \frac{\nu}{2} \left(\Gamma\left(\frac{3}{\nu}\right)\right)^{\frac{1}{2}} \left(\Gamma\left(\frac{1}{\nu}\right)\right)^{-\frac{3}{2}} \left(\frac{1}{\sigma_{t}}\right)$$
$$\exp\left\{-\left[\Gamma\left(\frac{3}{\nu}\right)/\Gamma\left(\frac{1}{\nu}\right)\right]^{\frac{\nu}{2}} |z_{t}|^{\nu}\right\}$$
(7)

with μ_t as the conditional mean, σ_t^2 as conditional variance, and $\Gamma(.)$ as the gamma function. The shape parameter ν indicated the degrees of freedom under the GED distribution. The GED distribution can be identical with the normal distribution for $\nu = 2$ and to the Laplace distribution for $\nu = 1$.

For 1 = 2, v = 2, and $v \to \infty$ degrees of freedom, the GED distribution converges to uniform density.

An estimation for the maximum of the log-likelihood function, which can also be considered a quasi-maximum likelihood function, is given as follows:

$$L(\theta) = \sum_{t=1}^{T} \log f\left(\mu_{t-1}, \sigma_{t-1}^{2}, \nu\right).$$
(8)

4. Data set

The dataset used will be described in the following paragraphs. For my analysis, both stock market data and weather data were collected, and the rationale for the specific data selection will be explained.

4.1. Stock market data

My investigation is based on companies of the Deutsche Aktienindex (DAX) 30, Mid-cap DAX (MDAX), and Small-cap DAX (SDAX). The TechDAX is not included, because recently, TechDAX companies were granted access to the MDAX and SDAX and are, thus, already incorporated in the given indices. The constitution of all three indices follows rules that are checked every quarter. Hence, for my analysis I use the companies which are contained in the first quarter of 2019. This is especially relevant for the MDAX and SDAX, which significantly change their composition regularly. A total of 121 companies is included, which is selected based on available stock market data. Data is retrieved from Datastream for a time horizon from 01 January 2009 to 28 February 2019 on a daily basis. This time frame was selected by reason of two arguments. First, it complies with time horizons used in extant research (Hirshleifer and Shumway, 2003; Shu, 2010) and second, it still comprises a relatively large set of companies, that is representative for the German business landscape. Moreover, a significant body of research has proven that the utilisation of the major index of a country serves as a representative dataset (Floros, 2008; Symeonidis et al., 2010). Therefore, the DAX, MDAX, and SDAX are suitable for an analysis of the German market.

Based on Datastream results, companies were sorted into nine industries, being Basic Materials, Consumer Cyclicals, Consumer Non-Cyclicals, Financials, Healthcare, Industrials, Telecommunication, Technology, and Utilities. The number of companies per sector is disparate which is explained through the historically shaped industrial landscape of Germany. It contains more companies in the sectors Industrials and Consumer Cyclicals than Telecommunication and Utilities as seen in Table 1. The appendix includes a list of companies in each sector for further reference.

To calculate the industry return indices, two option exists: First, the value-weighted average gives a better representation of the economy by putting more weight on companies with a larger number of outstanding shares and a higher market price per share than small companies. However, this

300

Table 1: Overview of the number of companies included per industry

Industry	Number of companies included
Basic Materials	14
Consumer Cyclicals	19
Consumer Non-Cyclicals	4
Financials	19
Healthcare	12
Industrials	30
Telecommunication	4
Technology	16
Utilities	3

method does not incorporate the quick growth of small enterprises and is exposed to the risk of overweighting a few incumbent companies. Second, an equally-valued approach, in turn, distributes market returns equally along with all included companies. This method benefits from higher exposure to small and medium-sized companies, which hold significant growth potential. A downturn is the missing realistic depiction of the stock-economy relation. In a considerable body of literature, both equally-and value-weighting methods have been used, but results for both cases were very similar (Cao and Wei, 2005; Kamstra et al., 2003; Saunders, 1993; Zheng et al., 2001). Zhu and Goetzmann (2003) used the equal weighted average only, because "average of the spread better reflects the average market maker's behavior" (p. 12). There are both arguments for and against the equally- and value-weighted strategy. For the calculation of the industry return indices, the equally-weighted method has been used.

The compounded returns of the industry indices have been used to receive a stationary series of stock market returns. Returns for each industry have been calculated as the change in the natural logarithm of the company price for the two successive days:

$$R_{t,j} = \ln(P_{t,j}) - \ln(P_{t-1,j})$$
(9)

 $P_{t,j}$ is the adjusted price at day t of industry j, incorporating stock splits and other corporate actions, apart from dividends. $P_{t-1,j}$ is the price of the previous day of industry j. This calculation is also in line with extant research (Floros, 2011; Sariannidis et al., 2016).

The data plots in Figure 2 to Figure 10 in the appendix depict the fluctuation of market returns for each of the nine industries. It can be noticed that returns are highly volatile and show evidence for volatility clustering.

4.2. Weather data

Weather data was obtained from the Climate Data Center (CDC) of the DWD. The platform offers a comprehensive set of historical data for several weather variables. For my analysis, historical weather data of Frankfurt am Main is used. The weather station with the station ID 1420 is located close to Frankfurt Airport at an altitude of 100m (8.5213, 50.059).

Weather data has been collected on a daily basis for the variables: the average temperature at an altitude of 2m in °C and precipitation height in mm per square meter.

It is also necessary to address the topic of location choice. Frankfurt is one of the most important financial centres in Europe and the most important one in Germany. With Xetra and Börse Frankfurt, it inherits the largest of the seven German stock exchange. The dispersion of market participants in Germany is, compared with the US, much smaller. Consequently, Frankfurt is a good proxy for the behaviour of traders located in Munich, Dusseldorf, and Stuttgart as well (Krämer and Runde, 1997). As the weather conditions are relatively moderate all over Germany, the weather of Frankfurt is utilised as a representative for the weather in Germany. Research has moreover proven that only the largest exchange in a country determines prices, as stated by Krämer and Runde. As a result, changes in market prices in Frankfurt are representative of other exchanges. This fact also holds through the effect of arbitrage. Saunders (1993) states that traders on Wall Street might have a severe impact on security prices because they are concerned about market prices. As they work at the same location every day, a mood variable might influence traders more than geographically spread market participants. This opinion is shared by Zhu and Goetzmann (2003) who include, next to market makers, also news providers and other agents who are physically located in the exchange hosting city. The behaviour of those market participants is conjectured to be responsible for a relation between weather and return. As traders are based in Frankfurt, it appears plausible to utilise this location. For my analysis, I will not further investigate e-trading, as this process is done automatically and is probably not influenced by traders' or investors' mood.

4.3. Descriptive statistics

Table 2 below presents Pearson correlation coefficients for the nine industry indices. One can observe a strong correlation between all industry returns, which are all significant at a 1% level. Correlation between industry returns implies an interconnectedness of industries. Industry indices seem to move in lockstep with each other. The correlation is highest for Basic Materials and Industrials with 0.8792*** and lowest for Consumer Non-Cyclicals and Telecommunication

Tab	le	2:	Corre	lation	matrix	for	stocl	k mar	ket	returns
-----	----	----	-------	--------	--------	-----	-------	-------	-----	---------

	Financials	Telecommu- nication	Technology	Consumer Non- Cyclicals	Industrials	Healthcare	Basic Materials	Consumer Cyclicals	Utilities
Financials	1								
Telecommunication	.6556***	1							
Technology	.7268***	.6188***	1						
Consumer Non-Cyclicals	.5985***	.5089***	.5459***	1					
Industrials	.8424***	.6822***	.8162***	.6380***	1				
Healthcare	.6666***	.6230***	.7060***	.5883***	.7429***	1			
Basic Materials	.7966***	.6309***	.7435***	.6002***	.8792***	.6768***	1		
Consumer Cyclicals	.7925***	.6551***	.7592***	.6053***	.8703***	.7054***	.8358***	1	
Utilities	.8312***	.6977***	.8090***	.6561***	.9181***	.8534***	.9374***	.9332***	1

Note: Asterisks indicate significance as follows: (*)10%, (**) 5%, (***) 1%. In parentheses: Standard errors.

with 0.5089***. Therefore, knowing that industry return's behaviour is quite similar, it will be interesting to investigate, whether the correlation with weather is identical for all industries.

Simple summary statistics of the market return and weather data are included in my analysis. Table 3 reports market returns. It provides insight into the nature of the distribution of returns that should be analysed before proceeding with further analysis. The first column shows the mean return, which varies across industries but is always slightly positive. Nevertheless, it is not possible to reject the null hypothesis that mean returns are not statistically different from zero. Standard deviation is relatively similar across all sectors. Basic Materials returns have the highest standard deviation and consequently the highest variance in returns. The return series of all industries are strongly skewed to negative returns and leptokurtic to the normal distribution. To evaluate the goodness-of-fit, a Jarque-Bera (JB) test is conducted, to analyse whether the skewness and kurtosis of the sample data match a normal distribution. Normally distributed data would have an expected kurtosis and skewness of zero. Deviation of a normal distribution will increase the JB value. In the selected sample, JB values range from 2,240.7704*** to 13,227.4538***. The Augmented Dickey-Fuller (ADF) test confirms that the market return variables are stationary, and therefore the hypothesis that a unit root exists is rejected.

Table 4 displays statistics of the weather data. Temperature is recorded in °C and precipitation in mm per square meter. Therefore, they do not have high explanatory value when compared with each other. However, skewness is valuable for interpretation. Temperature has a negative skew, just like returns. Consequently, it might be interesting to test for correlation between those variables and market returns. Further, precipitation has a substantial skewness of 4.3147 which indicates a strongly asymmetrical distribution. Its kurtosis is also extreme with 30.5830. Therefore, precipitation exhibits tail data that exceeds the tails of a normal distribution. This is a signal that Frankfurt is sometimes exposed to intense rainfall which could potentially influence investor's behaviour.

5. Hypothesis

Market returns are the result of a complex construct of influencing factors. Generally, the price is determined by supply and demand. However, this is just a theoretical approach, and also technical factors impact stock returns. Inflation, the economic strength of the market, and competitors influence investors' sentiments, attitude, and expectations. Further political and economic decisions can induce stock market activity. Moreover, the weather has often been argued to influence stock market returns. Therefore, keeping in mind that the trader's attitude can impact investment decisions, leads me to the hypotheses I want to test with the dataset and the methods discussed:

H0: Weather variables are not correlated with stock market returns. The investor is not significantly influenced by the weather.

H1: Weather variables are correlated with stock market returns. Investors are significantly influenced by weather.

The analysis of existing research implies a possible relationship between weather and stock market returns that would implicate to reject the null hypothesis. This will be discussed in the following section six "Empirical Results" in greater detail. Additionally, the purpose of this paper is not only to identify a potential relationship between stock market returns and weather, but also to investigate whether industries are affected differently by weather. This could result either in a difference in positive or a negative correlation with respect to each industry or in equally existent correlation.

H2: Industry market returns are equally influenced by weather.

H3: Industry market returns respond significantly different to weather.

6. Empirical results

The following paragraphs will investigate the implied hypotheses after the data and methodology used for my empir-

 Table 3: Descriptive statistics of stock market returns

Statistics	Financials	Telcommuni -cation	Technology	Consumer Non- Cyclicals	Industrials	Healthcare	Basic Materials	Consumer Cyclicals	Utilities
Mean	0.0005	0.0006	0.0008	0.0002	0.0005	0.0006	0.0002	0.0004	0.0004
Maximum	0.0728	0.0829	0.0725	0.0389	0.0631	0.0459	0.0803	0.0659	0.0520
Minimum	-0.0560	-0.1584	-0.0741	-0.0693	-0.0704	-0.0509	-0.0777	-0.0635	-0.0640
SD	0.0111	0.0142	0.0129	0.0100	0.0124	0.0105	0.0145	0.0118	0.0112
Skewness	-0.1965	-0.5350	-0.3526	-0.2723	-0.4000	-0.2827	-0.1475	-0.2639	-0.3041
Kurtosis	7.4156	10.8906	4.9359	5.2427	5.3707	4.4684	5.7647	5.8631	5.1275
Jarque-Bera	6,091.2391***	13,227.4538***	2,746.0117***	3,068.8088***	3,256.7562***	2,240.7704***	3,680.3264***	3,827.8807***	2,945.0084***
ADF	-47.9430***	-49.2850***	-48.6610***	-51.1070***	-46.9800***	-50.0000***	-48.8410***	-46.1710***	-47.9750***
Observations	2,651	2,651	2,651	2,651	2,651	2,651	2,651	2,651	2,651

Note: This table reports bin-test results for the equal-size sample. The sample covers the period 01.02.2009 to 28.02.2019 with a sample size of 2651 observations. All samples are in logarithmic first difference form and expressed as a percentage on a daily basis. Jarque-Bera test matches the skewness and kurtosis to the normal distribution. ADF is the Augmented Dicked-Fuller test for a unit root. Asterisks indicate significance as follows: (*) 10%, (**) 5%, (***) 1%.

Table 4: Descriptive statistics of weather data

Statistics	Temperature	Precipitation
Mean	11.1436	1.5558
Maximum	29.7000	50.2000
Minimum	-11.0000	0.0000
SD	7.5970	3.6679
Skewness	-0.0587	4.3147
Kurtosis	2.2562	30.5830
Observations	2,651	2,651

ical tests has been described, and it will summarise the outcomes of my analysis. The bin-test is discussed first. Afterwards, the results of the EGARCH model are described with the aim to find a relationship between weather and industry stock market returns. This will be done by testing two sets of variables. My study ends with a discussion of the outcomes and a further outlook.

6.1. Bin-test

The bin-test was performed for each industry separately, and results are depicted in Table 5. Generally, one can observe a negative correlation between return and temperature for seven of the nine industries. Thus, low temperatures are typically associated with greater returns than high temperatures. Financials and Telecommunication, however, show a positive correlation between returns and temperature. The percentage of positive returns is always higher for low temperatures when there is a negative correlation between returns and temperature. The z-score for differences in return means is only significant at the 5% level for Consumer Non-Cyclicals. The percentage of positive returns does not have any significant z-scores.

The results in Table 5 conform with the literature, as a negative correlation is reported in the majority of the sectors. High temperatures seem to be associated with lower returns and low temperatures with higher returns. This finding can be explained with psychological research, indicating that high temperature can cause apathy leading to lower risktaking. Consequently, apathy must strongly dominate aggression. On the opposite, aggression is dominant on days with low temperature leading to higher risk-taking which in turn leads to greater market returns. These conjectures are in line with the findings of Cao and Wei (2005).

The return behaviour in the Financials and Telecommunication sector raises the question for the underlying reason for this anomaly and could be explained with existing literature. Both high and low temperature can induce aggressive behaviour. Therefore, contradicting with the findings derived from the majority of industries, aggression might dominate apathy in certain occasions.

6.2. EGARCH

Table 6 and 7 report estimated parameters for equation (4) and (6) as outlined in section 3.2. Table 6 depicts estimates for the variance equations, having the underlying mean equation which is displayed in Table 7 with respective mean equations' estimates for each industry. The analysis has been conducted by using an EGARCH (1,1) model with GED. The reason for choosing this method is that the variance of stock-market returns varies over time and is dependent on past performance which has been further explained in section 3.2.

First, I will analyse the variance equation that offers insights into the distribution and behaviour of stock market returns. The persistence in conditional volatility, measured by γ_l , is highly significant and mostly has positive coefficient values close to one. Therefore, it is evident that GARCH effects are apparent in the German stock market and stock market

Table 5: Bin-test results

		Bin 1	Bin 2	Bin 3	Bin 4	z-score(4,1)
Financials	Return mean	0.00040	0.00053	0.00035	0.00062	0.20866
	% of pos. returns	0.53623	0.53800	0.52330	0.54912	0.26165
Telecommunication	Return mean	0.00034	0.00090	0.00040	0.00053	0.13921
	% of pos. returns	0.50000	0.52000	0.52061	0.50882	0.17844
Technology	Return mean	0.00124	0.00102	0.00076	0.00038	-0.68063
	% of pos. returns	0.52899	0.53100	0.53405	0.53652	0.15287
Consumer Non-	Return mean	0.00062	0.00018	0.00036	-0.00062	-1.26603
Cyclicals	% of pos. returns	0.55072	0.49700	0.50538	0.46851	-1.67111**
Industrials	Return mean	0.00071	0.00065	0.00035	0.00031	-0.34299
	% of pos. returns	0.57246	0.55100	0.52867	0.53401	-0.78498
Healthcare	Return mean	0.00073	0.00058	0.00073	0.00007	-0.71451
	% of pos. returns	0.55072	0.52100	0.54839	0.50630	-0.90264
Basic Materials	Return mean	0.00042	0.00027	0.00006	0.00023	-0.13717
	% of pos. returns	0.51449	0.48900	0.49552	0.51134	-0.06393
Consumer Cyclicals	Return mean	0.00021	0.00041	0.00064	-0.00014	-0.31193
	% of pos. returns	0.55797	0.51200	0.52419	0.51889	-0.79505
Utilities	Return mean	0.00045	0.00042	0.00048	0.00005	-0.38838
	% of pos. returns	0.56522	0.52700	0.53226	0.52393	-0.84120

Note: This table reports bin-test results for the equal-size sample. The sample covers the period 01.02.2009 to 28.02.2019 with a total of 2651 observations. The mean return and the percentage of positive returns are reported for each of the four bins. Z-scores are computed as follows: $z_{-} \operatorname{score}_{4,1}^{\text{mean}} = (\mu_4 - \mu_1) / \sqrt{\frac{\sigma_4^2}{n_4} + \frac{\sigma_1^2}{n_1}}$ and $z_{-} \operatorname{score}_{4,1}^{\text{frequence}} = (p_4 - p_1) / \sqrt{\frac{p_4(1-p_4)}{n_4} + \frac{p_1(1-p_1)}{n_1}}$ Asterisks indicate significance as follows: (*)10%, (**) 5%, (***) 1%.

Table 6: Estimates for the conditional variance

Coefficient	Financials	Telecommunication	Technology	Consumer Non- Cyclicals	Industrials	Healthcare	Basic Materials	Consumer Cyclicals	Utilities
$\overline{\alpha_0}$	-0.1907***	-3.2453***	-0.4332***	-0.2907***	-0.3863***	-0.6388***	-0.0926***	-0.2059***	-0.2836***
-	(0.0395)	(0.6492)	(0.0854)	(0.0812)	(0.0681)	(0.1178)	(0.0235)	(0.0476)	(0.0530)
α_1	0.1541***	0.3920***	0.1612***	0.1192***	0.1781***	0.1610***	0.0969***	0.1646***	0.1580***
	(0.0199)	(0.0728)	(0.0242)	(0.0228)	(0.0231)	(0.0276)	(0.0151)	(0.0214)	(0.0215)
ξo	-0.0937***	-0.1055***	-0.1004***	-0.0511***	-0.1343***	-0.1291***	-0.0898***	-0.0801***	-0.1204***
-	(0.0121)	(0.0447)	(0.0164)	(0.0136)	(0.0152)	(0.0188)	(0.0106)	(0.0141)	(0.0149)
γ1	0.9792***	0.6148***	0.9506***	0.9683***	0.9570***	0.9306***	0.9892***	0.9768***	0.9690***
	(0.0043)	(0.0775)	(0.0097)	(0.0088)	(0.0075)	(0.0127)	(0.0027)	(0.0052)	(0.0058)
ν	1.4203***	0.8780***	1.4870***	1.3938***	1.4283***	1.4531***	1.5798***	1.3506***	1.4867***
	(0.0551)	(0.0301)	(0.0583)	(0.0430)	(0.0591)	(0.0593)	(0.0674)	(0.0405)	(0.0622)
Log-likelihood	8629.7920	7778.1860	7986.2680	8602.1610	8250.7640	8528.4020	7836.6410	8370.9450	8491.2720
Adj. R ²	0.0039	0.0011	0.0025	-0.0002	0.0077	0.0003	0.0019	0.0107	0.0042

Note: This table presents parameter coefficients for the EGARCH(1,1) specification in order to estimate the conditional variance. The model is based on the conditional variance equation (6): $\ln(\sigma_t^2) = \alpha_0 + \sum_{l=1}^p \gamma_l \ln \sigma_{t-1}^2 + \sum_{l=1}^q \alpha_l [|z_{t-l}| - E|z_{t-1}|] + \sum_{l=1}^q \xi_l z_{t-l}$. Asterisks indicate significance as follows: (*)10%, (**) 5%, (***) 1%. In parentheses: Standard errors.

volatility is highly dependent on the previous period. Also, the magnitude effect α_l is highly significant, implicating that the size of shock influences the stock market response. However, the values are relatively small, which translates into a short memory of variance. ξ_l , measures the leverage effect. Its values are negative and highly significant for all industries, implicating that an asymmetrical response to past residuals

is observed. Thus, higher volatility is perceived for negative shocks than for positive shocks of the same magnitude. This observation indicates that investors are more influenced by negative compared to positive news. This fact could be translated to weather news, and consequently, especially high and low temperatures should be tested.

Further, the endogenously estimated shape parameter v

Coefficient	Financials	Telecommu -nication	Technology	Consumer Non- Cyclicals	Industrials	Healthcare	Basic Materials	Consumer Cyclicals	Utilities
b0	0.0010***	-0.0004	0.0011***	0.0005 *	0.0012***	0.0008***	0.0007 **	0.0008***	0.0009***
	(0.0003)	(0.0003)	(0.0004)	(0.0003)	(0.0003)	(0.0003)	(0.0003)	(0.0003)	(0.0003)
b1	0.0679***	0.0377***	0.0514***	-0.0111	0.0761***	0.0046	0.0517***	0.0941***	0.0555***
	(0.0191)	(0.0146)	(0.0195)	(0.0186)	(0.0196)	(0.0194)	(0.0197)	(0.0190)	(0.0196)
b2	-0.0000**	0.0001***	-0.0000	-0.0000*	-0.0001**	-0.0000	-0.0001**	-0.0000	-0.0000*
	(0.0000)	(0.0000)	(0.0000)	(0.0000)	(0.0000)	(0.0000)	(0.0000)	(0.0000)	(0.0000)
b3	0.0000	0.0001	0.0000	0.0000	0.0000	0.0000	-0.0000	-0.0000	-0.0000
	(0.0000)	(0.0000)	(0.0001)	(0.0000)	(0.0001)	(0.0000)	(0.0001)	(0.0000)	(0.0000)

Table 7: EGARCH parameter estimates

Note: The estimated parameters rely on the model, eq.(4): $R_{t,j} = b_0 + b_1 \sigma_t^2 R_{t-1,j} + b_2 \sigma_t^2 \text{Temp}_t + b_3 \sigma_t^2 \text{Prec}_t + \xi_{t,j}$. Parameters are estimated simultaneously with eq. (6): $\ln(\sigma_{t,j}^2) = \alpha_0 + \sum_{l=1}^p \gamma_l \ln \sigma_{t-1,j}^2 + \sum_{l=1}^q \alpha_l [|z_{t-1,j}| - E|z_{t-l,j}|] + \sum_{l=1}^q \xi_l z_{t-l,j}$ Asterisks indicate significance as follows: (*)10%, (**) 5%, (***) 1%. In parentheses: Standard errors.

shows that we can reject a normal or Laplace distribution for all industries except the Telecommunication sector. As $\nu < 2$, the tails are heavier than in a normal distribution. The distribution is leptokurtic and has positive excess kurtosis. Values are all significant at a 1% level, and range between 0.8780*** and 1.3506***. This implies a distribution that is more similar to Laplace than to a normal distribution. The distribution of Telecommunication is smaller than zero, while all other industry indices return values are greater than zero. The rationale behind this is that the sample size forming the industry index is very small and object to flaws.

The adjusted R^2 differs across industries. On average, the previous day's market return and weather data explain 0.36% of the market return. Weather explains only a minimal amount of market returns. In terms of magnitude and pattern, the adjusted R^2 is in line with research by Saunders (1993), Cao and Wei (2005) and Kamstra et al. (2003). Explanatory power is greatest for Consumer Cyclicals with 1.07% and smallest for Consumer Non-Cyclicals with 0.00% displaying that this model has no predictive value for this particular industry.

In the following, I will present results for the mean equation and analyse whether market returns are correlated with the weather variables, temperature and precipitation and if the relationships are significant.

Overall, the model affirms the existence of a relationship between weather and stock market returns. The lagged market return value is highly significant and positive for all industries. This finding goes hand in hand with the EGARCH model choice, as it proves the conjecture that past returns influence today's returns.

Temperature is also significantly correlated with industry index returns when considered together with precipitation. The coefficient parameter b2 is mostly negative and highly significant. Values are all close to zero and range from 0.00005*** for Telecommunication to -0.00006** for Basic Materials. The coefficients for Telecommunication and Financials are significant at the 1% level, the values for Industrials and Basic materials at the 5% level, and the coefficient for Consumer Non-Cyclicals and Utilities at the 10% level. Consequently, traders are influenced by temperature to a different extent depending on the industry. The small parameter coefficients come as no surprise and are in line with existing research, which frequently reports very small coefficient values for weather variables (Saunders, 1993). The fact that temperature mostly has a negative effect on market returns is in agreement with previous findings (Cao and Wei, 2005; Floros, 2011, 2008; Keef and Roush, 2002). To illustrate the economic significance inherited in the temperature effect, I will include an example that shows how daily returns react to a temperature shock of one standard deviation. The standard deviation for temperature measured in Frankfurt is 7.5970°C; the coefficient of temperature for Industrials is -0.00005. Consequently, the return impact of one standard deviation of temperature is 0.0004.

The one standard deviation impact, thus, is threequarters of the average daily return for the Industrials index.

Apart from temperature, precipitation is included in the model, which has a slightly positive, but no significant, effect on market returns. Hirshleifer and Shumway (2003) have also brought up evidence for the insignificance of rain on stock market returns.

In conclusion, according to this model, the null hypothesis, that weather variables and stock market returns are not correlated, can be rejected. My analysis provides evidence that traders are influenced by temperature and the past performance of stock prices. Further, the hypothesis that industry market returns are equally affected by weather is rejected as well. My outcomes reveal that significance in return depends on the industry.

6.3. Robustness test

My analysis has made use of an EGARCH model. As part of a thorough review, the robustness of the EGARCH model and regression must be examined. I will perform a robustness test for the EGARCH model, and for the regression analysis. First, a likelihood ratio (LR) test was conducted to verify the relative explanatory power of the EGARCH model. The underlying equation for the LR test is as follows:

$$LR = -2 * (L_0 - L_1) \tag{10}$$

LR is based on a chi-squared distribution, with L_0 being the likelihood function used under the null hypothesis and L_1 corresponding to the alternative hypothesis. The test, which is also called deviance, measures the difference in the LR of the nested model to more sophisticated models. The GARCH model is considered the nested model, L_0 and the alternative GARCH specifications are treated as L_1 . The smaller the absolute difference, the better the model fit. Table 8 illustrates the log likelihood values of each model and Table 9 the LR ratio test. LR is smallest for the EGARCH model, supporting my model choice. Additionally, for all industries, apart from Telecommunications, the EGARCH model yields the highest log-likelihood values. Therefore, this model has the best fit and verifies the robustness of the model choice. Secondly, I consider an alternative regression function for the EGARCH model which incorporates a calendar related anomaly to achieve a sufficient robustness test. I have chosen the Monday effect, also called the weekend effect, because it has been employed in a variety of research (Floros, 2008; Hirshleifer and Shumway, 2003). Studies have shown that the Friday closing price has an influence on the Monday opening price (Rogalski, 1984). Therefore, this effect has been added

to the mean equation (4). The Monday effect is incorporated as a dummy variable D_{Mon} , being one for the day Monday and zero otherwise. It yields the regression equation:

$$R_{t,j} = b_0 + b_1 R_{t-1,j} + b_2 Temp_t + b_3 \operatorname{Prec}_t + b_4 D_{Mon} + \xi_t$$
(11)

After controlling for this effect, the regression results should remain similar in order to achieve robustness in the utilised model. Indeed, temperature remains a significant variable after the dummy variable is added. Only Telecommunication does not report a significant correlation with weather anymore. Precipitation is still insignificant for all industries. Also, the day Monday, as the beginning of the trading week, does not influence stock market returns. The insignificance of the Monday effect contradicts the majority of existing research. However, it is in line with the findings of Krämer and Runde (1997), who analysed the German DAX returns and did not find any significant effect for specific days or the week.

In summary, the regression robustness test, my model is robust against the Monday effect. The temperature correlation coefficient remains negative for all industries and significant for the majority of sectors. Consequently, the EGARCH model has high informative value.

7. Discussion

From the empirical results, which have been discussed in detail, it can be conclude that temperature impacts stock market returns of particular industries. Observations from the underlying dataset identified Financials, Consumer Non-Cyclicals, Industrials, Basic Materials, and Utilities as industries that have a significant negative correlation between temperature and stock market performance. This chapter will elaborate on possible reasons for the observed relationship in the industries mentioned above and focuses on the analysis of economic impacts. It has been extensively discussed in section 2 that weather impacts mood, decisionmaking, and consequently investment decisions and risk- taking behaviour of traders. In addition to this effect, weather influences companies' economic performance. This effect is partially factored into the market price, but the extent to which investors are subconsciously influenced by weather effects is unknown. Therefore, I conjecture that both effects are subliminally integrated into the behaviour of traders and are therefore relevant aspects in the discussion.

Hot and cold temperature can influence several industrial areas. This paragraph will outline economic reasons, why low temperature leads to higher market returns and vice versa. The Industrials industry consists of companies in the logistics area, machinery, aviation, construction, and many more. Focussing on aviation, companies as Lufthansa are affected in their daily business by high temperatures. The take-off performance is negatively affected, because the air density decreases, the aeroplane will climb slower, leading to delays and requiring more fuel. The same problem exists for many logistics companies. In addition, extremely hot weather induces physical activity to be especially strenuous and is harmful to workers' health constitution and performance.

The Basic Material sector is very energy intensive. Especially in summer, machines need to be cooled down, which increases energy consumption and consequent energy costs.

In the Utilities industry, the primary energy sources are predominantly used for heating. Thus, low temperatures can be favourable because demand rises which in turn increases revenue.

For Consumer Non-Cyclicals, the interpretation is not as intuitive as for the other sectors. My sample includes industries active in the food producing sector (Suedzucker), growers and seed producers (KWS), personal care (Beiersdorf), and the Baywa group, which is active in agriculture, energy, and building materials. This sample, thus, still includes companies engaged in different fields. Therefore, my interpretation concentrates on the agricultural companies Baywa and KWS. The former benefits from the advantages of low temperatures for the same reasons as Utilities companies. This could outweigh the effect that increasing fruit and vegetable consumption is expected for higher temperatures. The latter might be influenced by hot spells implicating declining seed sales volume. The economic performance of Suedzucker might depend on the growth of the sugar beet. This argumentation is not exhaustive, and no reasonable explanations can be found for temperature implicationson the economics of Beiersdorf.

Turning to the Financial industry, my conjectures are

Table 8: Loglikelihood values

Statistics	Financials	Telecommu- nication	Technology	Consumer Non- Cyclicals	Industrials	Healthcare	Basic Materials	Consumer Cyclicals	Utilities
GJR	8463.963***	7764.910***	7904.216***	8562.111***	8092.710***	8457.401***	7647.490***	8211.179***	8326.612***
PARCH	8456.612***	7759.040***	7888.662***	8560.148***	8080.146***	8448.126***	7642.939***	8208.130***	8318.698***
GARCH	8607.276***	7836.726***	7972.032***	8588.664***	8216.340***	8506.574***	7805.412***	8355.436***	8458.939***
NARCH	8467.850***	7764.164***	7905.959***	8561.830***	8096.731***	8456.661***	7650.093***	8213.454***	8328.849***
EGARCH	8629.792***	7778.186***	7986.268***	8602.161***	8250.764***	8528.402***	7836.641***	8370.945***	8491.272***

Note: This table reports the log-likelihood values with respect to the selected GARCH model: GJR GARCH, Power ARCH (PARCH), Nonparametric ARCH (NARCH), EGARCH. In highlighted numbers indicate the greatest log-likelihood value per industry. Asterisks indicate significance as follows (*) 10%, (**) 5%, (***) 1%.

Table 9: Likelihood ratio statistics

Statistics	Financials	Telecommu- nication	Technology	Consumer Non- Cyclicals	Industrials	Healthcare	Basic Materials	Consumer Cyclicals	Utilities
GJR	-286.626***	-143.632***	-135.632***	-53.106***	-247.260***	-98.346***	-315.844***	-288.514***	-264.654***
PARCH	-301.328***	-155.372***	-166.740***	-57.032***	-272.388***	-116.896***	-324.946***	-294.612***	-280.482***
NARCH	-278.852***	-145.124***	-132.146***	-53.668***	-239.218***	-99.826***	-310.638***	-283.964***	-260.180***
EGARCH	45.032***	-117.080***	28.472***	26.994***	68.848***	43.656***	62.458***	31.018***	64.666***

Note: This table illustrates the LR statistic in order to test the significance of the models (GJR, PARCH, NARCH, and EGARCH) to the nested model GARCH. The LR statistic is calculated as follows: LR=-2*(L_0-L_1). Asterisks indicate significance as follows: (*) 10%, (**) 5%, (***) 1%.

Table 10: EGARCH robustness test includir	ng seasonality related anomaly variab	le
---	---------------------------------------	----

Coefficient	Financials	Telecommu- nication	Technology	Consumer Non- Cyclicals	Industrials	Healthcare	Basic Materials	Consumer Cyclicals	Utilities
b0	0.0010 ***	0.0005 ***	0.0010 ***	0.0004	0.0012 ***	0.0040 **	0.0006 *	0.0008 ***	0.0002 ***
	(0.0003)	(0.0003)	(0.0004)	(0.0003)	(0.0003)	(0.0194)	(0.0003)	(0.0003)	(0.0003)
b1	0.0679 ***	0.0520	0.0512 ***	-0.0125	0.0762 ***	-0.0000	0.0517 ***	0.0941 ***	0.0550 ***
	(0.0191)	(0.0176)	(0.0195)	(0.0186)	(0.0195)	(0.0000)	(0.0197)	(0.0190)	(0.0196)
b2	-0.0000**	-0.0000	-0.0000	-0.0000*	-0.0001**	0.0000	-0.0001**	-0.0000	-0.0000*
	(0.0000)	(0.0000)	(0.0000)	(0.0000)	(0.0000)	(0.0000)	(0.0000)	(0.0000)	(0.0000)
b3	0.0000	0.0000	0.0000	0.0000	0.0000	0.0006	-0.0000	-0.0000	-0.0000
	(0.0000)	(0.0001)	(0.0001)	(0.0000)	(0.0001)	(0.0004)	(0.0001)	(0.0000)	(0.0000)
b4	0.0001	0.0007	0.0004	0.0006	0.0001	0.0006	0.0002	0.0000	0.0002
	(0.0004)	(0.0005)	(0.0005)	(0.0004)	(0.0004)	(0.0003)	(0.0005)	(0.0004)	(0.0004)

Note: The estimated parameters rely on the model, eq. (11): $R_{t,j} = b_0 + b_1 R_{t-1,j} + b_2 \text{Temp}_t + b_3 \text{Prec}_t + b_4 D_{\text{Mon}} + \xi_{t,j}$. Parameters are estimated simultaneously with eq. (6): $\ln(\sigma_{t,j}^2) = \alpha_0 + \sum_{l=1}^p \gamma_l \ln \sigma_{t-1,j}^2 + \sum_{l=1}^q \alpha_l [|z_{t-l,j}| - E|z_{t-1,j}|] + \sum_{l=1}^q \xi_l z_{t-1,j}$. Asterisks indicate significance as follows: (*) 10%, (**) 5%, (***) 1%. In parentheses: Standard errors.

mainly based on the findings of Subak et al. (2000). By analysing the implications of the anomalously hot weather during summer in 1995 in the UK, they find that the recorded insurance claims for property damage through subsidence were much higher than for bursting pipes. Soil subsidence is a natural cause of lasting drought and heat. Furthermore, outdoor fires are sometimes caused by heatwaves, increasing the number of insurance cases. Consequently, hightemperature periods constitute a less economically profitable scenario for companies in this sector, than more moderate weather circumstances.

It has been previously reported in the literature that climate changes, in particular temperature, can impact the economic dimension in the Telecommunications sector (Adams and Steeves, 2014). Specifically, high temperatures lead to more need for I equipment cooling in base stations or exchanges, which can result in an escalating failure rate. Also, malfunction and reduced life span of network-and telecommunication equipment are possible. Energy demand increases as a result of intensified cooling activities and might lead to a rise in energy prices. These are, among others, reasons explaining why high temperatures are harmful to companies in the telecommunication sector.

Taken together, industry performance is impacted by climate conditions. However, the argumentation outlining why low temperatures are favourable for business performance and high temperatures harm the industrial segment is onesided, and it must be noted that the effects of climate can also be opposing. In other words, there are numerous unfavourable implications of cold weather.

Another relevant aspect of this topic is the psychological implication for investor behaviour. As a result of the economic impact I conjecture that these complex correlations between temperature and stock markets are triggered intuitively in market makers. Thus, on days with hotter temperatures traders subconsciously incorporate the environmental circumstances into their evaluation of the respective industries. In addition, they also consider these factors in their investment decisions.

8. Conclusion

My findings are essential to understand the implications of weather on stock market returns. The gained knowledge could be insightful for developing weather-based trading strategies. To derive a benefit from this strategy, transaction costs need to be very low because the correlation coefficient of stock market returns is minimal and consequently, economic gains are marginal (Hirshleifer and Shumway, 2003). Another finding of this paper is that it raises the general awareness that people are likely to be influenced by climate factors. If traders, or investors, know that their investment decisions might be impacted by the outside temperature, ill-conceived investments could be prevented.

8.1. Outlook and limitations

Additionally, seven caveats must be addressed that concern potential flaws in methodology and data choice. First, the investor location must be addressed. By choosing Frankfurt, I can depict the behaviour of traders. This does not include independent market makers or individual investors who are widely distributed in Germany and over the whole world (Sariannidis et al., 2016). If order-submitting investors are price setters, and investors' mood is affected by weather, Frankfurt is not a good proxy and can potentially deteriorate results. This limitation could be addressed by calculating a population-weighted temperature based on the greatest German cities (Cao and Wei, 2002), but as investors are not limited to Germany, this is not a sound solution.

Second, traders usually work in office buildings which are furnished with air cooling as well as heating systems. Indoor temperature levels are retained at a medium temperature of approximately 21°C. If the temperature does correlate with human behaviour and market returns, this occurrence is confined to the extent to which traders are exposed to the outside temperature, e.g., during their commute. Accordingly, it is "the psychological imprint of the extreme temperature that mediates people's mood and influences their behaviour" (Cao and Wei, 2002, p. 1562). Nevertheless, it remains questionable whether the exposure to the outside environment is significant enough to utilise temperature as a regression variable.

Third, other economy-related variables could be included in the regression model because it is evident, that weather, if at all, only explains a vanishing small portion of market return behaviour. Especially, variables related to global environmental emissions might affect stock market returns (Sariannidis et al., 2016).

Fourth, weather and the economy are dynamic, and climate anomalies are never alike. Therefore, it is not reasonable to draw implications out of a climate anomaly to a future period.

Fifth, the researchers Krämer and Runde (1997) claim that findings of Saunders could also be a result of a type one error due to statistical inference. This flaw could be limited by collecting data from a longer period of time than ten years. Further, they claim that the significance of a correlation between the variables weather and market returns is subject to data mining. Depending on the definition of good and bad weather, relationships can either be identified or dismissed. Zheng et al. (2001) also share this view. They claim that by studying historical stock market returns one can find significant relationships just by chance.

Sixth, existing research has mostly focused on global or national stock market indices that do not differentiate between industries. As my analysis has proven, correlations do only exist for specific industries. Thus, by combining all sectors in an index, the weather effect is diluted because significant and insignificant industries are combined. This fact could explain why researchers obtain diverging results and would challenge the outcome correctness of extant work.

Seventh, no psychological literature exists that directly analyses the impact of weather on investment behaviour. This lack of research has already been pointed out by Cao and Wei (2002). The conjunction is generally made by financial researchers that extrapolate from feelings and mood to investment activities. Consequently, it is questionable whether a direct link between weather and market returns exists. One needs to strictly differentiate between correlation and causation. My paper could only prove correlation which does not guarantee any causal relations. This limitation is based on potential endogeneity problems within the model implicating that an independent variable is potentially correlated with the error term. Three sources of endogeneity will be briefly discussed.

It is difficult to avoid omitted variables. Uncontrolled variables can be correlated with both independent and dependent variables or independent variables and the error term. The confounding variables lead to inconsistent estimates and biases. I cannot rule out the existence of omitted variables from my model. Another source of endogeneity is the simultaneity bias which means that the explanatory and dependent variable are jointly determined. This bias is unlikely as weather is not caused by stock market but by meteorological dynamics. Additionally, measurement errors of the dependent variable can result in inconsistent parameter estimates. Market returns can be calculated based on various methods. The prices and returns chosen in this model might be prone to errors. Instrumental variables, among other techniques, can be utilized to address the problem of endogeneity. However, completely exogenous instruments are very difficult to find.

All limitations suggest that more research is needed in this field, in particular, with respect to the analysis of weather implications on industry returns. Especially, this field should be evaluated on a global level as has been done in research that investigated the relationship of a variety of global indices with weather. Besides, further studies should incorporate a finer depiction of weather data and incorporate individual and institutional investors into the model. Hitherto, only weather and seasonal variables have been included in research. The literature is outdated to a great extent. On the one hand, one could argue, that volatility in weather does not change. On the other hand, global warming is increasingly getting public attention. Hence, it would be interesting to connect market returns not only with weather but include a climate change variable that depicts profound variations in weather trends and influences mood to a yet unknown extent. This procedure is also valuable, considering that climate change has already had major implications on industry standards and policies, as the automotive industry has illustrated (Williams et al., 2018).

8.2. Final remarks

In the psychological literature, it is well established that weather can influence people's mood and feelings. Mood, in particular, impacts people's decision-making behaviour. From a financial point of view, mood also impacts investment-decisions and risk assessment. The empirical literature has identified temperature as one of the most critical weather variables that affect human behaviour and mood: Low temperature can cause aggression, high temperate might induce apathy. These mood alternations can lead to either higher or lower risk attitude. A body of empirical literature has revealed a relationship between weather and market returns. Examples are sunshine (Saunders, 1993), temperature (Cao and Wei, 2005), length of the night (Kamstra et al., 2003), and many more.

It is examined if weather alternates mood and investment behaviour. This is done by analysing the relationship between market returns and the weather variables temperature and precipitation. For this, a Bin-test and an EGARCH model are applied. First, I hypothesise that weather variables are correlated with stock market returns. Market returns include the German indices DAX 30, MDAX, and SDAX and are separated by industries. I consider the second hypothesis: Industry returns respond significantly differently to the weather.

After the examination of the industry indices, I can provide evidence that traders are influenced by temperature and the past performance of stock prices. The correlation between temperature and stock market returns is negative and significant for six out of nine industries. Lagged returns are all significant and mostly positively correlated with stock market returns. Precipitation, on the other hand, is not significantly correlated with stock market returns. The significance in return depends on the industry which supports the hypothesis that industry market returns do not respond equally to weather effects. The industries Technology, Healthcare, and Consumer Cyclicals do not show a significant correlation. However, for the other six industries, a significant negative correlation between temperature and stock market returns was identified. This relationship is robust to changes in the underlying model and seasonal anomalies as the Monday effect.

All in all, temperature has a negative relationship with some industries. Future research should elaborate on the underlying reasons, why particular industries' market returns are affected by weather variables. Furthermore, studies on an industry-wide level should be extended to a global scale.

References

- Adams, P. and Steeves, J. Climate risks study for telecommunications and data center services Risks Study for Telecommunications and Data Center Service: Report prepared for the general service administration, 2014.
- Bless, H., Schwarz, N., and Kemmelmeier, M. Mood and Stereotyping: Affective States and the Use of General Knowledge Structures. *European Review of Social Psychology*, 7(1):63–93, 1996. ISSN 1046-3283. doi: 10.1080/14792779443000102.
- Cao, M. and Wei, J. Stock market returns: A temperature anomaly: Working Paper. 2002.
- Cao, M. and Wei, J. Stock market returns: A note on temperature anomaly. Journal of Banking & Finance, 29(6):1559–1573, 2005. ISSN 03784266. doi: 10.1016/j.jbankfin.2004.06.028.
- Chang, S., Chen, S., Chou, R. K., and Lin, Y. Weather and intraday patterns in stock returns and trading activity. *Journal of Banking & Finance*, 32 (9):1754–1766, 2008. ISSN 03784266. doi: 10.1016/j.jbankfin.2007. 12.007.
- Chang, T., Nieh, C.-C., Yang, M. J., and Yang, T.-Y. Are stock market returns related to the weather effects? Empirical evidence from Taiwan. *Physica A: Statistical Mechanics and its Applications*, 364:343–354, 2006. ISSN 03784371. doi: 10.1016/j.physa.2005.09.040.
- Cunningham, M. R. Weather, mood, and helping behavior: Quasi experiments with the sunshine samaritan. *Journal of Personality and Social Psychology*, 37(11):1947–1956, 1979. ISSN 0022-3514. doi: 10.1037/ 0022-3514.37.11.1947.
- Dichev, I. D. and Janes, T. D. Lunar cycle effects in stock returns. *The Journal* of Private Equity, 6(4), 2001. URL SSRN281665.
- Dowling, M. and Lucey, B. M. Weather, biorhythms, beliefs and stock returns—Some preliminary Irish evidence. *International Review of Financial Analysis*, 14(3):337–355, 2005. ISSN 10575219. doi: 10.1016/j.irfa .2004.10.003.
- Eagles, J. M. The relationship between mood and daily hours of sunlight in rapid cycling bipolar illness. *Biological Psychiatry*, 36(6):422–424, 1994.
- Floros, C. Stock market returns and the temperature effect: new evidence from Europe. *Applied Financial Economics Letters*, 4(6):461–467, 2008. ISSN 1744-6546. doi: 10.1080/17446540801998585.
- Floros, C. On the relationship between weather and stock market returns. *Studies in Economics and Finance*, 28(1):5–13, 2011. ISSN 1086-7376. doi: 10.1108/10867371111110525.
- Hirshleifer, D. and Shumway, T. Good Day Sunshine: Stock Returns and the Weather. *The Journal of Finance*, 58(3):1009–1032, 2003.
- Howarth, E. and Hoffman, M. S. A multidimensional approach to the relationship between mood and weather. *British Journal of Psychology*, 75 (1):15–23, 1984. ISSN 00071269. doi: 10.1111/j.2044-8295.1984.tb 02785.x.
- Imbery, F., Friedrich, K., Koppe, C., Janssen, W., Pfeifroth, U., Daßler, J., and Bissolli, P. 2018 wärmster Sommer im Norden und Osten Deutschlands, 2018.
- Isen, A. M. Missing in Action in the AIM: Positive Affect's Facilitation of Cognitive Flexibility, Innovation, and Problem Solving. *Psychological Inquiry*, 13(1):57–65, 2002.
- Kals, W. S. Your health, your moods, and the weather. New York, N.Y., Doubleday Books, New York, 1 edition, 1982.
- Kamstra, M. J., Kramer, L. A., and Levi, M. D. Losing Sleep at the Market: The Daylight Saving Anomaly. *American Economic Review*, 90(4):1105– 1011, 2000.
- Kamstra, M. J., Kramer, L. A., and Levi, M. D. Winter Blues: A SAD Stock Market Cycle. American Economic Review, 93(1):324–343, 2003.
- Kang, S. H., Jiang, Z., Lee, Y., and Yoon, S.-M. Weather effects on the returns and volatility of the Shanghai stock market. *Physica A: Statistical Mechanics and its Applications*, 389(1):91–99, 2010. ISSN 03784371. doi: 10.1016/j.physa.2009.09.010.
- Keef, S. P. and Roush, M. L. The Weather and Stock Returns in New Zealand. Quarterly Journal of Business and Economics, 41(1/2):61–79, 2002.
- Keef, S. P. and Roush, M. L. Influence of weather on New Zealand financial securities. Accounting and Finance, 45(3):415–437, 2005. ISSN 0810-5391. doi: 10.1111/j.1467-629x.2004.00135.x.
- Keef, S. P. and Roush, M. L. Daily weather effects on the returns of Australian stock indices. *Applied Financial Economics*, 17(3):173–184, 2007. ISSN 0960-3107. doi: 10.1080/09603100600592745.

Krämer, W. and Runde, R. Stocks and the weather: An exercise in data

mining or yet another capital market anomaly? *Empirical Economics*, 22 (4):637–641, 1997.

- Loughran, T. and Schultz, P. Weather, Stock Returns, and the Impact of Localized Trading Behavior. *The Journal of Financial and Quantitative Analysis*, 39(2):343–364, 2004.
- Lu, J. and Chou, R. K. Does the weather have impacts on returns and trading activities in order-driven stock markets? Evidence from China. *Journal* of *Empirical Finance*, 19(1):79–93, 2012. doi: 10.1016/j.jempfin.2011. 10.001.
- Mackie, D. M. and Worth, L. T., editors. Feeling good, but not thinking straight: The impact of positive mood on persuasion. Emotion and social judgment. Amsterdam, Pergamon/Elsevier, Oxford, 1991.
- Nelson, D. B. Conditional Heteroskedasticity in Asset Returns: a new approach. Econometrica The Econometric Society, 59(2):347–370, 1991.
- Pardo, A. and Valor, E. Spanish Stock Returns: Where is the Weather Effect? European Financial Management, 9(1):117–126, 2003.
- Parsons, A. G. The Association Between Daily Weather and Daily Shopping Patterns. Australasian Marketing Journal (AMJ), 9(2):78–84, 2001. ISSN 14413582. doi: 10.1016/S1441-3582(01)70177-2.
- Pilcher, J. J., Nadler, E., and Busch, C. Effects of hot and cold temperature exposure on performance: a meta-analytic review. *Ergonomics*, 45(10): 682–698, 2002. ISSN 0014-0139. doi: 10.1080/00140130210158419.
- Rind, B. Effect of Beliefs About Weather Conditions on Tipping. Journal of Applied Social Psychology, 26(2):137–147, 1996. ISSN 0021-9029. doi: 10.1111/j.1559-1816.1996.tb01842.x.
- Rogalski, R. New findings regarding day of the week returns over trading and non-trading periods. *Journal of Finance*, 39:1603–1614, 1984.
- Rotton, J. and Rosenberg, M. Lunar cycles and the stock market: Time-series analysis for environmental psychologists. 1984.
- Sariannidis, N., Giannarakis, G., and Partalidou, X. The effect of weather on the European stock market. *International Journal of Social Economics*, 43 (9):943–958, 2016. doi: 10.1108/IJSE-04-2015-0079.
- Saunders, E. D. Stock Prices and Wall Street Weather. The American Economic Review, 83(5):1337–1345, 1993.
- Schwarz, N. Feelings as information: Informational and motivational functions of affective states. *Handbook of motivation and cognition: Foundations of social behavior*, 2, 1990.
- Shu, H.-C. Investor mood and financial markets. Journal of Economic Behavior & Organization, 76(2):267–282, 2010. ISSN 01672681. doi: 10.1016/j.jebo.2010.06.004.
- Shu, H.-C. and Hung, M.-W. Effect of wind on stock market returns: evidence from European markets. *Applied Financial Economics*, 19(11):893–904, 2009. ISSN 0960-3107. doi: 10.1080/09603100802243766.
- Subak, S., Palutikof, J. P. d. Agnew, M., Watson, S. J., Bentham, C. G., Cannell, M. G. R., Hulme, M., McNally, S., Thornes, J. E., Waughray, and Woods, J. C. The Impact of the Anomalous Weather of 1995 on the U.K. Economy. *Climatic Change*, 44(1/2):1–26, 2000. ISSN 01650009. doi: 10.1023/A:1005566710285.
- Symeonidis, L., Daskalakis, G., and Markellos, R. N. Does the weather affect stock market volatility? *Finance Research Letters*, 7(4):214–223, 2010. ISSN 15446123. doi: 10.1016/j.frl.2010.05.004.
- Tietjen, G. H. and Kripke, D. F. Suicides in California (1968–1977): Absence of seasonality in Los Angeles and Sacramento counties. *Psychiatry Research*, 53(2):161–172, 1994. ISSN 01651781. doi: 10.1016/ 0165-1781(94)90107-4.
- Trombley, M. A. Stock Prices and Wall Street Weather: Additional Evidence. Quarterly Journal of Business and Economics, 36(3):11–21, 1997. URL https://www.jstor.org/stable/pdf/40473319.pdf?re freqid=excelsior%3A8adc329de54eefe6f55c56792d31c163.
- Watson, D. Emotions and social behavior. Mood and temperament. New York, N.Y., Guilford Press, New York, 2000.
- Williams, J., Ferguson, M., Petkov, M., and Wilkins, M. The effect of weather events on corporate warnings are gathering force, 2018.
- Wright, W. F. and Bower, G. H. Mood effects on subjective probability assessment. Organizational behavior and human decision processes, 52(2): 276–291, 1992.
- Yoon, S.-M. and Kang, S. H. Weather effects on returns: Evidence from the Korean stock market. *Physica A: Statistical Mechanics and its Applications*, 388(5):682–690, 2009. ISSN 03784371. doi: 10.1016/j.physa.2008.11. 017.
- Zheng, L., Yuan, K., and Zhu, Q. Are Investors Moonstruck? Lunar Phases

and Stock Returns. Journal of Empirical Finance, 13(1):1–23, 2001. doi: 10.2139/ssrn.283156.
Zhu, N. and Goetzmann, W. N. Rain or Shine: Where is the Weather Effect? European Financial Management, 11(5):559–578, 2003.